Author's response to reviews

Title: Date of origin of the SARS coronavirus strains

Authors:

Dr Hongchao Lu (lhc@ict.ac.cn)
Dr Yi Zhao (biozy@ict.ac.cn)
Jingfen Zhang (jzhang@ict.ac.cn)
Yuelan Wang (wyl_lisa@sina.com)
Dr Wei Li (liw@genomics.org.cn)
Xiaopeng Zhu (nimezhu@163.com)
Shiwei Sun (dwsun@ict.ac.cn)
Lun Cai (cailun@ict.ac.cn)
Jingyi Xu (xjy@ict.ac.cn)
Lunjiaang Ling (ling@sun5.ibp.ac.cn)
Dongbo Bu (bdb@ict.ac.cn)
Runsheng Chen (crs@sun5.ibp.ac.cn)

Version: 4 Date: 3 Feb 2004

Ms. Emma Parkin
Editorial Administrator,
BMC Journals,
E-mail: editorial@biomedcentral.com

Feb 3, 2004

Dear Ms. Emma Parkin,

We agree on changing the case of the title from 'Date of Origin of the SARS Coronavirus Strains' to 'Date of origin of the SARS coronavirus strains' and changed it in this revised version. And we also used the ENDNOTE to correct the format of the references according to your advice.

Hope that the version will be acceptable at last and published soon. If there is any problem about the paper, please no hesitated contact me.

Respectfully yours,
Dr. Hongchao Lu
Bioinformatics Research Group,
Institute Computing Technology,
Chinese Academy of Sciences
Beijing 100086, China
E-mail: lhc@ict.ac.cn